# Early Poverty Exposure Predicts Young Adult Educational Outcomes in China

Xiaoying Liu<sup>1,2</sup> and Emily Hannum<sup>1,2</sup>

- 1. Population Studies Center, University of Pennsylvania.
- 2. Center for the Study of Contemporary China, University of Pennsylvania.

Corresponding author: Xiaoying Liu, xiaoyliu@sas.upenn.edu, phone: (+1) 609-865-6491. Corresponding address: 580 McNeil Building, 3718 Locust Walk, Philadelphia, PA 19104-6298

Emily Hannum: hannumem@soc.upenn.edu, phone: (+1)215-898-9633. Address: 247 McNeil Building, 3718 Locust Walk, Philadelphia, PA 19104-6298

# ABSTRACT

Comparative research suggests that childhood poverty, and especially early childhood poverty, impedes educational attainment. With longitudinal data from China, we estimate hazard models of dropping out of school in young adulthood with two dynamic measures of childhood poverty: poverty spell indicators that distinguish poverty in early childhood, middle childhood, and adolescence; and poverty indices that measure the degree of poverty and distinguish chronic from transient poverty.

Four main results emerge: 1) Children who experience spells in poverty leave school at a higher rate, even after escaping poverty in later periods; 2) Transient poverty is more widespread, and shows a greater detrimental effect, than chronic poverty; 3) Early and middle-childhood poverty show greater detrimental effects than poverty in adolescence; and 4) Girls may be more susceptible than boys to early poverty, but among the never poor, girls may be educationally advantaged. We further test a possible mechanism: early nutrition poverty. We find that lower protein intake at an early stage of life is related to poorer educational outcomes in young adulthood.

#### JEL Classifications: I25, I32, O15

Key words: human capital, poverty, early childhood, middle childhood, gender, life cycle

# **1** INTRODUCTION

Household poverty has been linked to poorer educational access and attainment in many societies. For example, using asset indices developed from Demographic and Health Surveys from 35 countries, Filmer and Pritchett (1999) constructed a ranking of households within each country, and defined "poor" as the bottom 40 percent. Filmer and Pritchett found a prominent pattern in many countries in which the bulk of the deficit from universal basic education came from the poor. Huisman and Smits (2009) adopted the same approach for measuring wealth, but employed different cutoffs for designating poverty and analyzed enrollment at ages 8 to 11 in a combined sample of 222,853 in 30 countries from the Demographic and Health Surveys and the Pan Arab Project for Family Health of the League of Arab States. Huisman and Smits (2009) found that two indicators of socioeconomic status—parents' education and household wealth— remain strongly predictive even in specifications controlling for other family background and structural effects, along with district educational, demographic and economic characteristics. Mechanisms of the disadvantage may have to do with parental ability to pay for schooling, or the detrimental effect of poverty on the school functioning of children.

While cross-sectional associations between economic deprivation and educational outcomes are well-established, few studies in low and middle-income countries have sought to trace the implications of early poverty exposure to educational outcomes in young adulthood. Moreover, few studies in such contexts have sought to distinguish the impact of chronic versus transient poverty, and few have considered whether the timing of poverty matters, despite evidence from longitudinal research in developed countries suggesting that the timing and duration of poverty condition its impact on children's outcomes (Duncan and Magnuson 2013; Holmes and Kiernan 2013).

In this paper, using longitudinal data from China, we investigate links between early poverty exposure and later educational outcomes. Using a national poverty line adjusted for regional differences in prices, we adopt two approaches: 1) models that capture poverty spells, and 2) an analysis that distinguishes chronic from transient poverty. We define poverty measures for early childhood, middle childhood, and adolescence. We test these measures in hazard models of school leaving in young adulthood, and analyze their associations with the likelihood of completing compulsory school. In all analyses, we test whether poverty operates differently for girls and boys. The paper is organized as follows: the second section provides background of the research in poverty and child development; section three introduces poverty dynamics and its measurements; section four introduces data and sample selection; section five provides empirical analysis and main results; section six lists a few robustness checks and tests possible mechanisms; section seven concludes.

## **2 BACKGROUND**

Poverty may directly affect enrollment status through lack of ability to afford school fees or other direct or opportunity costs, in which case we might expect to see that poverty exposure during the ages that coincide with higher-cost stages of education might be most detrimental to youth outcomes. Or, poverty might affect preparation and readiness for school. Research in the United States and in low and middle income countries has indicated that household poverty, variously measured, is a risk-factor for many outcomes for children that could hinder capacity to function in school, including physical health and language and cognitive development (Grantham-McGregor et al. 2007; Engle and Black 2008; Walker et al. 2011; Yoshikawa, Aber, and Beardslee 2012, 273). Multiple developmental risks associated with poverty are likely to co-occur (Grantham-McGregor et al. 2007; Walker et al. 2011). For example, among 12 and 15 year olds in Andhra Pradesh, India, child deprivation is higher for chronically poor households across multiple indicators of health, education, and quality of life (Singh and Sarkar 2014).

The potential impact of household poverty on school preparedness is also suggested in the comparative academic achievement literature. Using PISA reading score data from 15 year-olds in the United States and the 13 OECD countries that scored higher than the United States, Ladd (2012, 209) showed that reading achievement was strongly correlated with the Economic, Social and Cultural Status (ESCS) socioeconomic index in all 14 countries. Using books in the home as a proxy for household economic status, Hanushek and Woessmann (2011) concluded that the association of achievement with this indicator of economic status was robust across the board in countries analyzed in the TIMSS-95 and TIMSS-Repeat, net of controls for age, gender, family status, immigration and parent immigration status, and test cycle (Hanushek and Woessmann 2011:118, Figure 2.2). Chiu's (Chiu 2007, 515) analysis of science achievement in the PISA data also showed a significant coefficient for books.

Research suggests that the timing of impoverishment may also be important for children (Boyden and Cooper 2007). Exposure to poverty during early childhood, middle childhood, or late childhood has different implications on children's development, and possibly their later outcomes. For example, Hoddinott and Kinsey (2001) find that children aged 12-24 months lose growth in the aftermath of a drought in Zimbabwe, while older children were unaffected in growth. To the extent that ability to pay fees or forego opportunity costs is the main mechanism by which poverty impacts schooling, we might expect that in school systems where private costs tend to increase with stage of schooling and where children's potential contribution to the family coffers increases with age, that poverty exposure during middle or later childhood would be most detrimental.

At the same time, some cross-national evidence suggests that earlier poverty experiences can be more damaging than later ones for children's developmental outcomes (Bird 2007; Alderman 2011; Engle 2012), especially that poverty-induced growth faltering before 24 months is "irreversible" (Victora et al. 2008, 2010, Hoddinott et al. 2008). Although recent studies have found that some catch-up growth occurs after two years, still the first two years are very important (Crookston et al. 2013, Prentice et al. 2013, Schott et al. 2013). Assessing literature for the United States available through the middle of the 1990s, Brooks-Gunn and Duncan (1997) concluded that children who experience poverty during their preschool and early school years have lower rates of school completion than children and adolescents who experience poverty only in later years. Duncan and his colleagues (1998) analyzed data from the Panel Study of Income Dynamics and showed that family economic conditions in early childhood had the greatest impact on completed schooling. Guo's (1998) analysis using National Longitudinal Survey of Youth data from the United States distinguished development tests from achievement and showed that long-term poverty had substantial influences on both, but with different time patterns. The younger ages were a more crucial period for the development of cognitive ability than early adolescence, but poverty experienced in adolescence appeared more influential on adolescent achievement than poverty experienced earlier in life. One possible mechanism might be through nutrition. A randomly assigned early-life nutritional supplementation program finds that poor nutrition during early life does cause poor adult cognitive skills (Maluccio et al. 2009). Timing aside, the duration of poverty is also important. In the United Kingdom, persistently poor children have more disadvantageous developmental contexts than children in poverty for shorter periods, and they exhibit worse developmental outcomes (Holmes and Kiernan 2013).

To our knowledge, with some notable exceptions (e.g., van der Gaag, Pells, and Knowles 2012; Pells 2011), few studies in low and middle income countries have investigated the dynamic implications of childhood poverty exposure for outcomes in young adulthood. This gap in research also exists in China, despite the fact that China has experienced rapid economic growth during the last three decades such that many middle-aged adults experienced poverty during early childhood and poverty alleviation in later periods. This situation provides a valuable natural opportunity for researchers to study the relationship between early poverty exposure and later outcomes, and whether the impact is irreversible. In this paper, we use the case of China to address the question, how are poverty dynamics in childhood linked to subsequent educational outcomes in young adulthood? We distinguish long and short-term poverty, and address the question of whether there are critical periods for poverty impact on education. We also consider whether there are gender differences in the impact of childhood poverty on educational outcomes. Finally, we test a possible mechanism: undernutrition.

## **3 POVERTY DYNAMICS AND MEASUREMENTS**

Poverty dynamics describe the movements into and out of poverty (Baulch and Hoddinott, 2000). It focuses on the temporal component of poverty, which is difficult to capture by a static measurement. A temporarily poor household suffering from a random negative income shock may not be permanently poor, and the impact of this poverty spell on children's development would be different from the one of the permanent poverty, which usually results in worse outcomes. Moreover, not only the length of the current poverty spell matters, but also the pattern and degree of poverty throughout childhood does: whether it consists of a series of intermittent spells of poverty, a single long spell of moderate poverty, or a short spell of extreme poverty (Bradbury, Jenkins and Micklewright, 2001). As said earlier, timing of poverty spells is also an important issue. It is important to capture all these different dimensions of poverty dynamics in the measurements.

Multiple conceptual approaches and measurement strategies have been applied to the study of poverty (for a review, see Grusky and Kanbur 2006; see also Christiaensen and Shorrocks 2012). Commonly, poverty is measured by comparing resources to needs, and families or individuals are considered poor if they fall below some threshold (J. E. Foster 1998, 335). Three steps are needed to quantify the scope of poverty (World Bank 2011; J. E. Foster 1998, 335): selecting welfare measures, constructing thresholds, and tallying the resulting data.

The first step is to define the relevant welfare measure (World Bank 2011). Income is a common choice, the use of which may be justified on the grounds that income is essential in societies with market economies (Borgeraas and Dahl 2010). Consumption is another common choice, dictated by the same logic; some have argued that consumption is a more appropriate longer-run measurement of household welfare than income because households can to a degree shift resources over time and smooth consumption more than income (Deaton 1997). A different approach, popularized by Sen and others, focuses on capabilities or endowments (Sen 1999; Sen 2006; Bourguignon 2006; Nussbaum 2006). The capabilities approach suggests that poverty should be defined in terms of individuals' capabilities to choose their own lives. This approach is associated with efforts to define poverty as failure to reach minimally acceptable levels in multiple dimensions—different monetary and nonmonetary attributes necessary for functioning (for example, see Bourguignon and Chakravarty 2003; Thorbecke 2005; Bourguignon 2006; and Nussbaum 2006; Marlier and Atkinson 2010; for a critical discussion of child-focused multidimensional indicators of poverty, see Dercon 2012).

Second, one has to select a poverty line –a threshold below which a given household or individual will be classified as poor (World Bank 2011). This line can be defined in absolute, relative, or subjective terms. One of the earliest conceptualizations of poverty was that of absolute poverty, commonly defined as lacking the income necessary to acquire sufficient material resources to satisfy basic physiological needs (Borgeraas and Dahl 2010, 73). Often, absolute poverty thresholds are developed for some initial period based on the cost of a "nutritional basket" considered minimal for the healthy survival of a typical family, to which a provision is added for non-food needs (J. E. Foster 1998, 336; World Bank 2011). The threshold is then carried forward, with adjustments for inflation. In national settings where non-trivial shares of the population survive at the margins, much poverty estimation work continues to rely on an absolute definition (World Bank 2011; see also Brady 2003, in passing). <sup>1</sup>

Third, one has to select an index to tally results, to be used in reporting for either a whole population or a subpopulation (World Bank 2011). A common choice is a trio of measures quantifying the incidence, depth, and severity of poverty developed by Foster, Greer, and Thorbecke (1984), which are the poverty headcount index, poverty gap index, and squared poverty gap index (e.g., see Haughton and Khandker 2009).<sup>2</sup> The headcount index (P<sub>0</sub>) measures the proportion of the population that is poor; the poverty gap index (P<sub>1</sub>) measures the extent to which individuals fall below the poverty line (the poverty gaps) as a proportion of the poverty gap index (also known as the poverty severity index, P2) averages the squares of the poverty gaps relative to the poverty line (Haughton and Khandker 2009, 67).

In this paper, we employ income as the primary welfare measure for defining poverty. This is a compromise as consumption is not measured in the CHNS. Household income was conceptualized as the sum of all sources of income and revenue minus production expenditures, and was constructed as the sum from nine potential sources -- business, farming, fishing, gardening, livestock, non-retirement wages, retirement income, subsidies, and other income. The income value in each wave was then inflated to 2009 Yuan current values. Per capita household income was calculated based on the inflated household income and the number of permanent residents in the household<sup>3</sup>. We adopted the absolute poverty

<sup>&</sup>lt;sup>1</sup> Additional approaches have also emerged. For example, in contrast to poverty lines demarcating absolute deprivation, relative poverty lines are defined in relation to the overall distribution of income or consumption in a country (Brady 2003; World Bank 2011). In this framework, poverty is not just about absolute levels of deprivation, but about deprivation relative to prevailing norms of income or consumption (or, by extension, some other indicator). Another approach has been to conceptualize the cutoff subjectively, in terms of "the amount of income it takes to barely get by" or other related questions (Yoshikawa, Aber, and Beardslee 2012, 273).

<sup>&</sup>lt;sup>2</sup> However, many other measures have been proposed (for example, see Brady 2003; Marlier and Atkinson 2010; Haughton and Khandker 2009).

<sup>&</sup>lt;sup>3</sup> Although equivalence scale is desired in calculating income measurement for poverty definition, there are no

line developed by Ravallion and Chen (2007), which is 1200 RMB per capita for urban households, and 850 RMB per capita for rural households, at 2002 prices. We then adjusted this line by national urban- and rural- specific consumer price indices (CPI) and calculated poverty indices based on the adjusted lines for each survey wave.

We consider poverty across time points among individuals, rather than seeking to report indices at a given time point for a population. We do so by incorporating transient versus chronic poverty, and the timing of poverty, into our measurement approach. We adopt two general approaches to calculating poverty for individuals, across time. The first approach adopted in this paper is a simple indicator of exposure to spells of poverty (dropping to the poverty line or lower) during different stages of childhood. We constructed a series of binary variables indicating whether one's household's income was above the poverty line in each specific time spell. Therefore, the longitudinal poverty status of each household was described by a vector of dichotomous variables with 1 indicating that the income in a period was below the poverty line and 0 otherwise. Examining poverty spells allows us to identify whether there are critical periods for poverty associated with children's outcomes. However, this approach, especially with income as welfare measurement for defining poverty, is well known to suffer from measurement error, due both to the intrinsic difficulties in measuring all variables and the special problems posed by recall and imputing values for own production in the case of expenditure and of estimating costs and revenues for agricultural and non-farm enterprises in the case of income (Baulch and Hoddinott 2000).

This concern leads us to adopt a second approach, in which we distinguish chronic and transient poverty by calculating chronic and transient poverty indices (see Ravallion and Chen 2007). For individual *i*, the *chronic poverty index* (*CP<sub>i</sub>*) measures the degree of long-term poverty by comparing long term average income  $(\bar{y}_i)$  against the poverty line (PL). Long-term average income is calculated as the total observed income for household *i* at each time *t*, divided by the total times observed *T*. Mathematically, it can be written as:  $CP_i = \frac{1}{T} \sum_{t=1}^{T} \left[ \left( 1 - \frac{\bar{y}_i}{PL} \right)_{+} \right]^2$ , where  $\bar{y}_i = \frac{1}{T} \sum_{t=1}^{T} y_{it}$  and  $f_+ = \max(f, 0)$ . If long term average income is less than the poverty line, the index is 0. If long term average income is less than the individual's income as a fraction of the poverty line. The index approaches 1 as an individual's income approaches 0.

The transient poverty index  $(TP_i)$ , in contrast, measures the degree of transient poverty that is due to inter-temporal variability in income. Mathematically, it is the difference between the squared poverty gap (Foster, Greer, and Thorbecke 1984),  $SPG_i = \frac{1}{T} \sum_{t=1}^{T} \left[ \left( 1 - \frac{1}{T} \sum_{t=1}^{T} \sum_{t=1}^{T} \left[ \left( 1 - \frac{1}{T} \sum_{t=1}^{T} \left[ \left( 1 - \frac{1}{T} \sum_{t=1}^{T} \sum_{t=1}^{T} \left[ \left( 1 - \frac{1}{T} \sum_{t=1}^{T} \sum$ 

standard scale indices that apply to Chinese households except an attempt by Chen (2006). For this reason, we use the per capita income which may overestimate the extent of poverty as some households likely fall under poverty line by simply having a new-born baby. We incorporated this possibility by controlling for the number of children in the household in the later empirical estimations.

 $\left(\frac{y_{it}}{PL}\right)_{+}\right]^{2}$ , and the chronic poverty index defined above:  $TP_{i} = SPG_{i} - CP_{i}$ . We consider overall exposure to transient poverty, and the timing of transient poverty during childhood.

Both of these approaches measure poverty dynamics. They are also complementary to each other as they measure poverty in different dimensions. The poverty spell approach emphasizes the timing of poverty, while the chronic and transient poverty approach measures both the degree and persistence of poverty. For example, if a household's income was observed to be below the poverty line in all waves, its chronic poverty index would be positive and the poverty spell vector would include all ones. But if a household fell under the poverty line in one of all waves observed, with long term average income above the poverty line, its chronic poverty index would be zero and transient poverty index would be positive, and it would have only one poverty spell indicating the period under poverty.

# 4 DATA AND SAMPLE SELECTION

## 4.1 DATA

The analyses in this study utilize data from the China Health and Nutrition Survey (CHNS) (see Popkin 1994). The CHNS is a longitudinal panel study designed to examine how the social and economic transformation of Chinese society is affecting the health and nutritional status of its population. The survey used a multistage, random cluster process to draw a sample of about 4400 households with a total of 26,000 individuals in nine provinces that vary substantially in geography, economic development, public resources, and health indicators. Detailed community data were also collected for food markets, health facilities, and other social policies. The study commenced in 1989, and revisited the panel every 2 to 4 years, in 1991, 1993, 1997, 2000, 2004, 2006, 2009 and 2011.

Although the CHNS is often criticized as not being representative of the Chinese population, its longitudinal features, especially the focus on health and nutrition, makes it unique in its application to this study. As many households were followed in multiple waves of the surveys, we are able to trace the whole growth periods of a small cohort of children into their young adulthood, with observations of their family economic status, school enrollment, education attainment. This is the feature that is only available in CHNS, among all the social and economic surveys from China, including the China Family Panel Studies (CFPS), Chinese Household Income Project (CHIP), and the Urban and Rural Household Surveys in China (UHS and RHS). Moreover, children's cohort data were not collected until recently (e.g., the China-Anhui Birth Cohort Study from 2008) or were established among children beyond the early childhood phase (e.g., the Gansu Survey of Children and Families, established in 2000). Thus, the CHNS is currently the only option to study this topic in China.

## 4.2 SAMPLE SELECTION

In order to model the poverty dynamics exposed by children, we would ideally need to observe the whole growth periods of a cohort of children. But this is impossible given the nonconsecutive design of CHNS, and that the CHNS doesn't follow the same cohorts over periods. Individuals were observed at different age points, and therefore we have to find common spells in which most children were observed to secure a sufficiently large sample size. Therefore, we define our analytic sample as the one that consists of all individuals observed at least once during each of the four periods: fetus  $\sim 3$ ,  $4 \sim 9$ , and  $10 \sim 15$ , and 16 + years old. This may not coincide with the biological clock that marks different growing needs recognized by pediatricians, such as 0 to 2 years for infants/babies, 2 to 5 years for toddlers/preschoolers, 6 to 12 years for school age children, and 13-18 for adolescents/teenagers. But this is the best that we can do given the data structure. There are 1311 individuals who satisfy this criteria, among whom 1103 were observed with household income in each above period, and only 606 were observed with schooling information after 16 years of age. Furthermore, 544 individuals were observed with parents' education and other control variables. In addition, 41 individuals were observed dropping out of school before they reached 16 years old. We assume they never returned to formal schooling after dropping out, and also include them in our sample. Table 1 shows definitions and descriptive statistics for some basic variables used in the analysis. And for purpose of looking at sample selection issue, we compare this analytic sample against the data pool which include all children born in the same birth-year periods as those in the analytic sample (1985-1995), but who may not necessarily be observed in all those growth periods (see column 1 in Table 1). In the later robustness check, we also define an extended sample which include both analytic sample and additional 473 individuals who were observed during each of the growth periods before 16, but have missing education information beyond  $16^{4}$ .

<sup>&</sup>lt;sup>4</sup> 705 individuals were in the roster in each period without schooling information after 16 years old. 179 of them were no longer household members since last observed before 16 years old, and another 526 did not live in the household though they were still household members (235 went to school and lived away from home; 12 served in military; 266 sought employment elsewhere; 1 went abroad; and 12 left for other reasons). For all these people, we do not observe them dropping out of school when last observed before 16 years old; and when they were away from home, they didn't answer the Child/Adult survey so that no schooling information was updated. For 473 of these individuals, we can observe their previous education information (before 16 years old) as well as other variables and therefore include them in our robustness check analysis with the complete sample.

	(1)		(2)		(3)			
	Data p	ool	p diff between	Analytic s	ample	p diff between	Extended	sample
	(n=206	52)	(1) and (2)	(n=58	5)	(2) and (3)	(n=47	3)
	mean	s.d.		mean	s.d.		mean	s.d.
Male	0.53		0.05	0.57		0.02	0.51	
Urban	0.26		0.01	0.21		0.00	0.12	
Birthyear	1989.31		0	1988.25	(0.10)	0.03	1988.55	(0.10)
South	0.47		0	0.54		0.06	0.59	
Long term average income (yuan)	14106.3	(279.69)	0	16233.87	(461.76)	0.13	15212.69	(485.26)
Chronic poverty index				0.01	(0.03)	0.91	0.01	(0.00)
Transient poverty index				0.06	(0.09)	0.90	0.06	(0.00)
Mother's education (%)								
Illiterate	17.36		0.04	21.03		0.01	28.12	
Primary school	23.28		0.88	23.59		0.37	26	
Secondary school	37.97		0.51	39.49		0	31.08	
High school	15.03		0	15.9		0.62	14.8	
College	6.35		0	2.39		0.43	1.69	
Father's education (%)								
Illiterate	6.35		0.48	7.18		0.6	8.03	
Primary school	19.11		0.94	18.97		0.91	19.24	
Secondary school	46.12		0.76	46.84		0.7	47.99	
High school	19.84		0.5	27.01		0.4	24.74	
College	8.58		0	3.93		0.92	3.81	

Table 1. Descriptive statistics of the analytic sample and comparisons with other samples

Note: Data pool includes all children born between 1985 and 1995 with basic information on geography, income, and parents' education, but are not observed during each growth period as defined. Extended sample includes individuals with observations in each growth period, but missing education information beyond 16 years old.

The statistics show that the analytic sample differs from the data pool in various dimensions: the analytic sample is more male, older, more rural, and more southern, and has less educated parents, especially in terms of higher education. The households in the analytic sample also have higher long term average income than those in the data pool, which, however, may not have many observations to calculate long term average income. Compared to the extended sample, the analytic sample also includes more boys, but is less rural, is less from southern China, and has more educated mothers. However, these two samples don't differ significantly in long-term average income, or poverty indices, which alleviates concern about biased outcomes due to selectivity of the sample.

We further plot the survival curves of schooling for different samples: the analytic sample, the extended sample and the pool of children within the same birth periods (see Figure 1). In the analytic sample, over 97% children complete primary school (grade 6) and continue to secondary school, while over 90% of children complete compulsory school (grade 9), only 60% move forward to high school and only 25% of children enter higher education after completing high school (grade 12). The analytic sample has a higher dropout rate than the other two samples, especially at compulsory school and high school (grade 12). This partly could be explained by having more elder cohorts in the analytic sample. Moreover, all three samples show girls' advantage beyond compulsory school.



**Figure 1. Kaplan-Meier survival curves of schooling for different samples, by gender.** The extended sample in the right top panel includes both analytic sample and the "extended sample" in Table 1. And the bottom left panel includes both analytic sample and the "data pool" in Table 1.

Figure 2 shows the poverty dynamics during children's developmental periods in our analytic sample. The number in brackets denotes the number of observations in each scenario. The blue level stands for the infant and toddler stage (fetus-3 years old); the orange level stands for the child stage (4-9 years old); and the green level stands for the adolescent stage (10-15 years old). The chart shows among the analytic sample of 588 children, 280 were exposed to poverty before reaching four years of age. Out of these 280, 144 individuals continued to be under the poverty line during the child stage of 4-9 years old, while the other 136 children were out of poverty during the same age stage. Finally, only 65 of those who were poor at childhood stayed poor during the adolescent stage of 10-15 years old. And 35 of those 136 who were initially poor before 4 and moved out of poverty during 4-9 years old fell back into poverty during the adolescent stage of 10-15 years old. Overall, 61 children were exposed to chronic poverty (with positive chronic poverty index), while 391 were exposed to transient poverty by the definitions above.



**Figure 2.** The poverty dynamics of the households for the sample children during growth periods. The blocks on each level (represented by different colors) show the distribution of the households under/above poverty line in each life-cycle stage. Blue level stands for infant and toddler stage (fetus-3 years old), orange level for child stage (4-9 years old), and green level for adolescent stage (10-15 years old).

These two types of poverty dynamics measurement are found consistent with each other. In our analytic sample, there are 65 children living under the poverty line during all three growth periods, while there are only 61 children who have a positive chronic poverty index calculated based on the long term average income. On the contrary, 392 children have ever experienced poverty during their childhood, and 391 have positive transient poverty index based on our calculation. Figure 3 displays the distributions of chronic and transient poverty indices, which clearly shows the prevalence of transient poverty relative to chronic poverty. Moreover, from the poverty dynamics diagram, we see that more children experienced poverty in early childhood than in later childhood. Although a minority of children fell into poverty in later childhood, many more escaped poverty in later childhood. This general movement is consistent with China's economic growth since 1980s. And that transient poverty is much more common than the chronic poverty is also consistent with this general trend.



Figure 3. The distributions of chronic and transient poverty indices during childhood.

# **5** Empirical analysis and main results

## **5.1 Empirical methods**

As some individuals did not finish schooling in the last wave observed in our data, attained education is a right censored variable. For this reason, a hazard duration model was used to address the censoring problem. More specifically, we adopted the Cox proportional hazard model and evaluated the association between the hazard of *dropping out of school* and poverty during childhood. We defined the outcome variable as the number of completed years of formal education in regular school. This outcome variable was measured at young adulthood, i.e., 16 years old or older, in the last observed survey. For those who haven't

finished formal schooling in the last survey, the highest grade attained so far is used with a flag indicating still being in school. In all specifications, we control for child's gender, parents' education levels, a quadratic function of age at which the education outcome is measured, and the number of children in the households. In addition, to control for community effects that reflect the supply side of education, we also include an urbanization index<sup>5</sup> of the community during childhood, and the community education index. These two indices measure the extent of urbanization and the accessibility of public services and the average level of education in a community, with higher scores indicating higher degree of urbanization and better education, respectively. For each life-cycle stage, if one child was observed in more than one wave, we took the multi-wave averages of these two indices within the same stage. We defined the strata by province, urban/rural area and birth year, i.e. for each birth year cohort from the rural/urban part of same province, the baseline hazard is assumed to be the same. In order to distinguish different associations of poverty for boys and girls, we included interaction terms between poverty variables and gender in models. Long term average income was also included as a control variable in some specifications with poverty spells. By doing so, we are able to compare the education outcomes of children who have the same long term average income during childhood, but were exposed to poverty at different time spells. This strategy intends to identify critical periods of children's development associated with poverty, without being confounded by income effects.

## 5.2 POVERTY SPELLS

Table 2 shows models of school-leaving with poverty spells and long-term income. Model 1 shows a specification with only long-term income and gender, and shows a significant protective effect of long-term income. It shows that if there were 10% increase in long term average income, children would be 2.88% less likely to drop out of school. This is a relatively "pure" household income effect, as the community urbanization index and education index were both controlled. Overall, there is no significant difference between boys and girls.

Model 2 shows results for poverty spells alone. Here, poverty exposure at ages 4 to 9 is associated with greater likelihood of leaving school. Poverty exposure at ages 10 to 15, when children would be making the transition to more expensive levels of education, shows no effect, net of earlier poverty. Model 3 contains both long-term income and poverty spells. Net of long-term income, no poverty exposure at any period is significant. Models 4 and 5 test gender interactions with poverty exposure, without and with long-term income, respectively. Early childhood poverty exposure is associated with significant increases in the risk of school-leaving for girls, but not for boys, while middle-childhood poverty exposure is

<sup>&</sup>lt;sup>5</sup> There are 12 components defining the urbanization index: population density, economic activity, traditional markets, modern markets, transportation infrastructure, sanitation, communications, housing, education, density, health infrastructure, and social services (see Jones-Smith and Popkin 2010).

associated with increases in the risk of dropping out of school for both girls and boys, although the associations with boys are smaller

Dependent variables: hazard ratio of dropping out of formal schooling								
	(1)	(2)	(3)	(4)	(5)			
(log) long time income mean	-0.292***		-0.275*		-0.264*			
	(0.112)		(0.151)		(0.153)			
poor at age<4 <sup>a</sup>		0.0848	0.0314	0.345**	0.294*			
		(0.107)	(0.115)	(0.154)	(0.164)			
poor at age [4,10) <sup>b</sup>		0.264**	0.188	0.339*	0.267			
		(0.110)	(0.117)	(0.177)	(0.186)			
poor at age [10,16) <sup>c</sup>		-0.00695	-0.13	-0.162	-0.256			
		(0.124)	(0.139)	(0.175)	(0.181)			
male * poor at age<4 <sup>d</sup>				-0.428**	-0.427**			
				(0.205)	(0.209)			
male * poor at age [4,10) <sup>e</sup>				-0.111	-0.113			
				(0.221)	(0.226)			
male * poor at age [10,16) <sup>f</sup>				0.252	0.211			
				(0.232)	(0.239)			
Male	0.0488	0.056	0.0519	0.276	0.286			
	(0.100)	(0.102)	(0.102)	(0.187)	(0.186)			
Prob > F test of (a + d=0)				0.063	0.111			
Prob > F test of (b + e=0)				0.045	0.228			
Prob > F test of (c + f=0)				0.539	0.365			
Log pseudolikelihood	-577.083	-576.936	-575.919	-575.317	-574.392			

Table	2.	<b>Poverty</b>	spells an	nd edı	ucation	attainment	at adulthoo	od (n=585)

Note: The table shows the survival analysis on finished schooling in adulthood by fitting Cox proportional hazard model with censoring. The hazards are stratified by province, urban/rural area and birth years. All specifications control for parents' education level dummies, quadratic function of age, number of children under 16 in the household, urbanization index of the community during the whole childhood, and urbanization index for education. Superscripts **a**, **b**,..., **j** stand for the coefficients of each above poverty indices and interaction terms between poverty indices and gender. Robust standard errors clustered at community level are in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

## 5.3 CHRONIC AND TRANSIENT POVERTY

Table 3 shows results for analyses of school leaving using chronic and transient poverty measures. Model 1 shows a baseline specification that includes only gender and chronic and transient poverty measures, along with other controls stated in the table note. In this specification, transient poverty, but not chronic poverty, shows a significant effect for the subsequent hazard of school leaving. Model 2 adds a gender interaction to the model, and shows that the significant effect of transient poverty is less pronounced for males than for females. In other words, girls are particularly vulnerable to transient poverty, although both males and females were more likely to drop out of school if facing transient poverty.

Model 3 allows the transient poverty index value to vary by age, and shows that transient poverty at ages 0 to 3 and 4 to 9 are highly significant. For a 0.1 increase in the transient poverty index during early childhood, the hazard of dropping out of school increases by 5.54%<sup>6</sup>, while for a 0.1 increase in the transient poverty index during middle childhood, the hazard of dropping out of school increases by 4.99%. Finally, model 4 tests the interaction between age-specific transient poverty and gender. In this specification, transient poverty is most detrimental at the earliest period of childhood, and the detrimental effect is much stronger amongst girls than boys. All specifications show a male disadvantage among the non poor, although this disadvantage is not statistically significant.

<sup>&</sup>lt;sup>6</sup> The hazard ratio is the exponential of the coefficient in hazard model.  $5.54\% = (\exp(0.1*0.539)-1)*100\%$ .

Dependent variables: hazard ratio of dropping out of formal schooling							
	(1)	(2)	(3)	(4)			
Chronic poverty index <sup>a</sup>	-1.19	-4.75	-1.541	-5.171			
	(1.908)	(5.613)	(1.921)	(5.627)			
Transient poverty index <sup>b</sup>	2.196***	3.134***					
	(0.598)	(0.777)					
Chronic poverty* male <sup>c</sup>		4.72		4.682			
		(5.978)		(5.938)			
Transient poverty * male <sup>d</sup>		-1.828**					
		(0.904)					
Transient poverty index at age [0,4) <sup>e</sup>			0.539**	1.394***			
			(0.211)	(0.329)			
Transient poverty index at age [4,10) <sup>f</sup>			0.487**	0.129			
			(0.210)	(0.354)			
Transient poverty index at age [10,16) <sup>g</sup>			0.218	0.601*			
			(0.214)	(0.336)			
Transient poverty index at age [0,4) * male <sup>h</sup>				-1.367***			
				(0.393)			
Transient poverty index at age [4,10) * male <sup>i</sup>				0.666			
				(0.439)			
Transient poverty index at age [10,16) * male <sup>j</sup>				-0.492			
				(0.469)			
Male	0.0545	0.168	0.0439	0.192			
	(0.100)	(0.138)	(0.101)	(0.145)			
Prob > F test of (a+c=0)		0.700		0.639			
Prob > F  test of  (b+d=0)		0.000		0.000			
Prob > F test of (e+n=0)				0.000			
Prob > F  test of  (1+1=0)				0.008			
riob > r lest of (8+J-0)				0.183			
Log pseudolikelihood	-575.195dis	-574.188	-575.644	-571.126			

Table 3. Poverty	persistence and	l education	attainment	at adulthood	(n=585)
	T				· · · · /

Note: The table shows the survival analysis on finished schooling in adulthood by fitting Cox proportional hazard model with censoring. The hazards are stratified by province, urban/rural area and birth years. All specifications control for parents' education level dummies, quadratic function of age, number of children under 16 in the household at each growth stage, urbanization index of the community during the whole childhood, and urbanization index for education. Superscripts **a**, **b**,..., **j** stand for the coefficients of each above poverty indices and interaction terms between poverty indices and gender. Robust standard errors clustered at community level are in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

# **6** ROBUSTNESS CHECK

## 6.1 COMPULSORY SCHOOL COMPLETION

Although survival analysis shows early childhood poverty exposure is negatively associated with children's future education attainment, it doesn't tell us about the probability of completing compulsory school, which may be more related with poverty and more relevant for poverty alleviation policies. In this section, we check whether the patterns we find in the early section remain in models with compulsory school completion as an outcome variable. The probit estimation results are listed in Table 4a and 4b, separately. Column (3) in Table 4a implies that those who were exposed to early childhood poverty are 4.74% less likely to complete compulsory school, compared to those who had the same long term average income during childhood, but were not exposed to poverty before 4 years old. Long-term average income is not significantly associated with compulsory school completion, which indirectly indicates the effectiveness of compulsory school enforcement in China. A similar pattern exists for a poverty persistence specification, as shown in Table 4b. Transient poverty at early childhood appears detrimental to girls' chances of completing compulsory school, but not to boys. Again, chronic poverty doesn't seem to predict lower chances of completing compulsory school.

Probit model on the probability of completing compulsory school									
(1) (2) (3) (4) (5)									
-0.079		0.001		-0.015					
(0.171)		(0.224)		(0.228)					
	-0.374**	-0.374**	-0.668***	-0.671***					
	(0.183)	(0.180)	(0.252)	(0.254)					
	0.03	0.03	0.08	0.08					
	(0.198)	(0.204)	(0.269)	(0.279)					
	0.430**	0.430*	0.44	0.44					
	(0.193)	(0.235)	(0.287)	(0.296)					
			0.57	0.57					
			(0.353)	(0.355)					
			-0.07	-0.07					
			(0.342)	(0.342)					
			-0.04	-0.04					
			(0.406)	(0.415)					
0.354**	0.357**	0.357**	0.044	0.043					
(0.161)	(0.167)	(0.169)	(0.294)	(0.295)					
			0.029	0.028					
			0.955	0.962					
			0.093	0.200					
-136 042	-132 646	-132 6/6	-131 583	-131 581					
	(1) -0.079 (0.171) 0.354** (0.161) -136.042	(1) (2)   -0.079 -0.374**   (0.171) -0.374**   (0.183) 0.03   (0.198) 0.430**   (0.193) 0.430**   (0.193) 0.430**   (0.161) (0.167)   -136.042 -132.646	(1) (2) (3)   -0.079 0.001   (0.171) (0.224)   -0.374** -0.374**   (0.183) (0.180)   0.03 0.03   (0.198) (0.204)   0.430** 0.430*   (0.193) (0.235)   0.354** 0.357** 0.357**   (0.161) (0.167) (0.169)	(1)   (2)   (3)   (4)     -0.079   0.001   (0.171)   (0.224)     -0.374**   -0.374**   -0.668***   (0.183)   (0.180)   (0.252)     0.03   0.03   0.03   0.08   (0.198)   (0.269)     0.430**   0.430*   0.444   (0.193)   (0.235)   (0.287)     0.430**   0.430*   0.430*   0.444   (0.353)   -0.07     0.353)   (0.287)   0.57   (0.353)   -0.07   (0.342)   -0.04   (0.406)   0.354**   0.357**   0.357**   0.044   (0.161)   (0.167)   (0.169)   (0.294)   0.294   0.029   0.955   0.093   -132.646   -131.583					

Table 4a. Poverty spells and compulsory school completion (n=573)

Note: All models include the same control variables as in Table 2, plus province and urban/rural sector dummies and birth year. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

Probit model on the probability of completing compulsory school								
	(1)	(2)	(3)	(4)				
Chronic poverty index <sup>a</sup>	3.317	1.807	3.145	0.797				
	(2.08)	(4.56)	(2.11)	(4.48)				
Transient poverty index <sup>b</sup>	-0.0249	-0.749						
	(0.99)	(1.36)						
Chronic poverty* male <sup>c</sup>		1.522		2.22				
		(5.09)		(5.24)				
Transient poverty * male <sup>d</sup>		1.632						
		(1.98)						
Transient poverty index at age [0,4) <sup>e</sup>			-0.051	-1.053**				
			(0.32)	(0.48)				
Transient poverty index at age [4,10) <sup>f</sup>			-0.0415	0.563				
			(0.40)	(0.61)				
Transient poverty index at age [10,16) <sup>g</sup>			0.104	0.016				
			(0.37)	(0.60)				
Transient poverty index at age [0,4) * male <sup>n</sup>				2.274***				
				(0.64)				
I ransient poverty index at age [4,10) * male'				-1.042				
				(0.73)				
Transient poverty index at age [10,16) * male				0.2				
Mala	0.995**	0.911	0 999**	(0.78)				
male	(0.335)	(0.211)	0.336	(0.131)				
	(0.102)	(0.211)	(0.100)	(0.217)				
Prob > F test of (a+c=0)		0.280		0.495				
Prob > F test of (b+d=0)		0.712						
Prob > F test of (e+h=0)				0.001				
Prob > F test of (f+i=0)				0.347				
Prob > F test of (g+j=0)				0.894				
Log pseudolikelihood	-140.799	-140.399	-140.752	-136.093				

Table 4b.	Poverty	persistence and	d compulsory	school co	mpletion (	n=573)
		1				,

Note: All models include the same control variables as in Table 2, plus province and urban/rural sector dummies and birth year. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1.

### 6.2 SAMPLE ATTRITION

The above analysis was conducted based on the analytic sample in which all individuals were observed and had education information from infant and toddler stage (0-3) through adult period (16+). This definition, however, raises an issue of sample attrition in that we ruled out those individuals who were not observed through adulthood. To check whether our analysis was biased by a selective sample, we checked robustness of analysis on a larger sample, consisting of the original sample and another 473 cases who were included in the roster in each analysis period and observed with their education information only before 16 years old, as well as other control variables (see footnote 3). We tested both specifications with this larger sample, and the results are in Appendix Table A1 and Table A2. Although the two groups were distinct in many dimensions as shown in Table 1, and the excluded group in the original analysis left home at earlier age, the quality and quantity of the empirical results didn't change much by including the second group. Poverty exposure in early childhood still has adverse effects on one's schooling. The only significant difference is that non-poor boys disadvantage in education attainment becomes significant in the last two models of both specifications.

## **6.3 POSSIBLE MECHANISM**

As mentioned in the background section, poverty influences education through different channels: nutrition, and family environment such as books and other stimulation that may affect children's preparedness and readiness in school; and the ability to pay for tuition. The CHNS is uniquely suited to investigating one potential mechanism: early nutrition. In the CHNS, nutrition intake is measured by a 3-day food diary of the household. Food diary items have been translated into three broad nutrition categories: carbohydrates, protein and fat, which can be used to calculate total calorie intake. Individual intake is calculated by the recalled proportion of each food taken. To make the measures comparable across different age groups, we standardize the nutrition intakes by removing the age trend. We here run the hazard model of dropping out of school with each category of nutrition intake during each growth period, with or without controlling for income. The other control variables remain the same as noted earlier. The main results are summarized in Table 5.

Among three macro nutrition intakes, protein intakes in early childhood and adolescence are negatively associated with the hazard of dropping out of school, while carbohydrates intake during the middle childhood (between 4 and 9) are positively associated with the hazard of dropping out of school. One may argue these associations are simply reflecting the associations with income, as richer households are more likely to consume protein rich foods and less carbohydrates. This concern can be alleviated by controlling for incomes in the model (see the lower panel of Table 5). Including incomes in the model reduces the coefficients on nutrition, suggesting nutrition partly captures the income effect. But the associations of protein intake remain significantly negative even after controlling for income. Moreover, we can also argue that the above associations are likely to be causal, as fat intake, which is highly associated with income in the sample (p<0.05 for all growth periods), does not show any associations with schooling duration in either panel. We find no significant differences between boys and girls in the associations with nutrition. These results suggest a possible mechanism that early poverty is associated with later educational outcomes through early undernutrition.

	prote	ein	carbohy	carbohydrates		fat	
age<4a	-0.263**	-0.288*	-0.226*	-0.269	0.00541	-0.028	
	(0.113)	(0.148)	(0.128)	(0.170)	(0.073)	(0.120)	
age[4,10) <sup>b</sup>	0.137	0.128	0.257**	0.346*	0.073	-0.0385	
	(0.132)	(0.217)	(0.121)	(0.180)	(0.095)	(0.136)	
age[10,16) <sup>c</sup>	-0.301**	-0.165	-0.199	-0.0447	-0.0776	-0.0223	
	(0.147)	(0.289)	(0.150)	(0.173)	(0.072)	(0.138)	
male*age<4d		0.0611		0.0759		0.0536	
		(0.225)		(0.218)		(0.129)	
male*age[4,10) <sup>e</sup>		0.00186		-0.164		0.184	
		(0.254)		(0.186)		(0.144)	
male*age[10,16) <sup>f</sup>		-0.204		-0.353		-0.079	
		(0.320)		(0.274)		(0.179)	
control for income at each							
period	no	no	no	no	no	no	
Prob > F test of $(a + d=0)$		0.065		0.182		0.909	
Prob > F lest of (b + e=0)		0.624		0.109		0.268	
P(0) > F(est 0) (C + 1-0)		0.058		0.215		0.529	
	prote	ein	carbohy	drates	fa	t	
age<4a	<b>prot</b> e-0.246**	ein -0.292**	<b>carbohy</b> -0.211*	drates -0.24	<b>fa</b> 0.0306	t 0.00173	
age<4a	-0.246** (0.113)	ein -0.292** (0.146)	<b>carbohy</b> -0.211* (0.125)	drates -0.24 (0.174)	<b>fa</b> 0.0306 (0.073)	t 0.00173 (0.120)	
age<4a age[4,10) <sup>b</sup>	prote -0.246** (0.113) 0.134	ein -0.292** (0.146) 0.0955	carbohy -0.211* (0.125) 0.239*	drates -0.24 (0.174) 0.317*	fa 0.0306 (0.073) 0.127	t 0.00173 (0.120) 0.00662	
age<4a age[4,10) <sup>b</sup>	prote -0.246** (0.113) 0.134 (0.133)	ein -0.292** (0.146) 0.0955 (0.219)	carbohy -0.211* (0.125) 0.239* (0.122)	-0.24 (0.174) 0.317* (0.180)	fa 0.0306 (0.073) 0.127 (0.093)	t 0.00173 (0.120) 0.00662 (0.140)	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup>	prote -0.246** (0.113) 0.134 (0.133) -0.253*	ein -0.292** (0.146) 0.0955 (0.219) -0.157	carbohy -0.211* (0.125) 0.239* (0.122) -0.21	drates -0.24 (0.174) 0.317* (0.180) -0.0727	fa 0.0306 (0.073) 0.127 (0.093) -0.0677	t 0.00173 (0.120) 0.00662 (0.140) -0.0356	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup>	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289)	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates     -0.24     (0.174)     0.317*     (0.180)     -0.0727     (0.185)	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144)	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224)	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates     -0.24     (0.174)     0.317*     (0.180)     -0.0727     (0.185)     0.0483     (0.221)	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135)	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d male*age[4,10) <sup>e</sup>	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d male*age[4,10) <sup>e</sup>	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249)	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates     -0.24     (0.174)     0.317*     (0.180)     -0.0727     (0.185)     0.0483     (0.221)     -0.144     (0.184)	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144)	
age<4a age $[4,10)^{b}$ age $[10,16)^{c}$ male*age<4d male*age $[4,10)^{e}$ male*age $[10,16)^{f}$	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249) -0.142	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144 (0.184) -0.316	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144) -0.0418	
age<4a age $[4,10)^{b}$ age $[10,16)^{c}$ male*age<4d male*age $[4,10)^{e}$ male*age $[10,16)^{f}$	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249) -0.142 (0.327)	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144 (0.184) -0.316 (0.273)	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144) -0.0418 (0.184)	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d male*age[4,10) <sup>e</sup> male*age[10,16) <sup>f</sup> control for income at each	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249) -0.142 (0.327)	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144 (0.184) -0.316 (0.273)	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144) -0.0418 (0.184)	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d male*age[4,10) <sup>e</sup> male*age[10,16) <sup>f</sup> control for income at each period	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146) yes	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249) -0.142 (0.327) yes	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144 (0.184) -0.316 (0.273) yes	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069)	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144) -0.0418 (0.184) yes	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d male*age[4,10) <sup>e</sup> male*age[10,16) <sup>f</sup> control for income at each period Prob > F test of (a + d=0)	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146) yes	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249) -0.142 (0.327) yes 0.076	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144 (0.184) -0.316 (0.273) yes 0.227	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069) yes	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144) -0.0418 (0.184) yes 0.868	
age<4a age[4,10) <sup>b</sup> age[10,16) <sup>c</sup> male*age<4d male*age[4,10) <sup>e</sup> male*age[10,16) <sup>f</sup> control for income at each period Prob > F test of (a + d=0) Prob > F test of (b + e=0)	prote -0.246** (0.113) 0.134 (0.133) -0.253* (0.146)	ein -0.292** (0.146) 0.0955 (0.219) -0.157 (0.289) 0.0991 (0.224) 0.0425 (0.249) -0.142 (0.327) yes 0.076 0.643	carbohy -0.211* (0.125) 0.239* (0.122) -0.21 (0.156)	drates -0.24 (0.174) 0.317* (0.180) -0.0727 (0.185) 0.0483 (0.221) -0.144 (0.184) -0.316 (0.273) yes 0.227 0.154	fa 0.0306 (0.073) 0.127 (0.093) -0.0677 (0.069) yes	t 0.00173 (0.120) 0.00662 (0.140) -0.0356 (0.144) 0.044 (0.135) 0.196 (0.144) -0.0418 (0.184) yes 0.868 0.098	

Table 5. Three macro nutrition intakes during growth periods and education attainment at adulthood(n=490)

Note: The table shows the survival analysis on finished schooling in adulthood by fitting Cox proportional hazard model with censoring. The hazards are stratified by province, urban/rural area and birthyears. Macro nutrition intake are standardized against age by taking the residuals after fitting polynomial trend function of age. All specifications control for parents' education level dummies, quadratic function of age, number of children under 16 in the household at each growth stage, urbanization index of the community during the whole childhood, and urbanization index for education. Robust standard errors clustered at community level are in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

# 7 SUMMARY AND CONCLUSIONS

In summary, results show that long-term income significantly predicts educational outcomes in adulthood, but even net of long-term income, spells in poverty shape educational trajectories. Transient poverty, as measured here, shows greater detrimental impact on children's educational attainment than chronic poverty<sup>7</sup>. Moreover, early and middle-childhood poverty exposure appears more detrimental to outcomes than does poverty exposure in the 10 to 16 age range. Further, there is some suggestion that girls' education may be more susceptible than boys' to early poverty exposure, but among the never poor, girls may be advantaged. These latter findings are consistent with some prior work in China suggesting that girls' education has been vulnerable to household or community poverty (Emily Hannum 2003; Cherng and Hannum 2013), but showing an emerging gender parity and even female advantage at some stages in education, in China (Cherng and Hannum 2013; Davis et al. 2007; Emily Hannum 2003; Wu and Zhang 2010) and around the world (Grant and Behrman 2010). Finally, protein intake in early childhood and adolescence is linked to educational outcomes in young adulthood, suggesting a possible mechanism of impact.

By focusing on children's outcomes in young adulthood, this paper adds a life-cycle perspective to a literature that considers differences in nature of transient and chronic poverty in middle and low-income countries (Bayudan-Dacuycuy and Lim 2014; Adepoju 2012), and adds a new case study to the literature on the longitudinal implications of childhood poverty. Two main insights emerge. First, while underscoring the importance of long-term economic status for children's human capital acquisition, this study also highlights that transient poverty—which may be more unexpected than chronic poverty--also matters. In the income and poverty spell models, even net of long-term income, spells in poverty shape educational trajectories. Moreover, in the chronic and transient poverty models, transient poverty, as measured here, showed greater detrimental impact on children's educational attainment than chronic poverty. It is likely that families adapt differently to expected, long-

<sup>&</sup>lt;sup>7</sup> One shouldn't over-interpret the insignificant chronic poverty coefficients, as the variation in chronic poverty index is much smaller than the variation in the transient poverty index. The chronic poverty index is dominated by zeros as only 61 out of 585 sample children were exposed to chronic poverty before they reached 16 years old. Moreover, this study has one drawback with the data that we use to measure chronic poverty - the nonconsecutive time series of CHNS does not allow measurement of household income every year. As an outcome, the transient poverty may actually capture some part of chronic poverty as one wave change of income actually reflects changes over a few years.

term economic deprivation then to the kinds of shocks that might cause sudden poverty spells, such as job loss, catastrophic health care costs, or sudden crop loss. Children's education bears the marks of these short-term shocks. This finding is not unique from our study. Scott also finds that transitory shocks can have long term consequences in Chile (Scott, 2000).

Second, the estimates suggest that the timing of poverty matters: early and middlechildhood poverty exposure appears more detrimental to outcomes than does poverty exposure in the 10 to 16 age range. This finding is somewhat surprising in the China context, if one assumes that ability to pay is the primary mechanism, given that the costs of schooling have typically been considerably higher at the middle and high school levels for families. Instead, the patterning of impact of childhood poverty exposure suggests that a mechanism other than just family ability to pay may be at work, and is more consistent with the theory that economic deprivation in early childhood creates disparities in school readiness (e.g., Duncan and Magnuson 2013). Our further analysis that shows an effect of early nutrition intake, mainly protein, on later educational outcomes suggest a possible mechanism. This finding seems consistent with early child development theory that emphasizes the importance of in-utero and within 2 years environment on future development (Royer 2009. Barreca 2010, Black et al. 2007) Education is a cumulative process, and children who experience resource deprivation early may be at a substantial disadvantage. In the United States, Gershoff and her colleagues (2007) analyzed a national sample of 21,255 kindergarteners from the Early Childhood Longitudinal Study, and as income increased, parental investments and resources for the child increased, which enhanced academic skills. At the same time, higher income was associated with reduced material hardship and stress, and fewer child behavior problems. Cross-national research indicates a robust relationship between indicators of household economic status and achievement (e.g., Ladd 2012; Hanushek and Woessmann 2011; Chiu 2007).

This paper has several limitations. First, the estimated relationship between poverty and education attainment reflects association, not causality. Although taking care of the endogeneity of income is not impossible by examining the unexpected shocks of income, it is not easy with the current data set to get a meaningful sample size. Moreover, we believe the associations can still be interesting, especially with the findings about the timing of poverty. As far as we know, this paper is the first one focusing on the relationship between early childhood poverty and later outcomes in China. Though methodologically weak, it may provoke more research in this area. The second limitation in this study is in sample attrition. The analysis sample used in this paper can be quite selective by the requirement of households staying in the study for over 16 years. However, this is a common problem with any longitudinal studies, especially the ones covering the long life span of subjects. Several studies of some other prospective birth cohort data sets have found that such attrition in terms of observed characteristics did not affect substantially multivariate estimates (Behrman et al. 2009, Hoddinott et al. 2008, Maluccio et al. 2009). Off course, one needs to be cautious in interpreting the estimation results. Third, poverty indices are not calculated perfectly as data collection is not continuous in years and therefore transitory poverty index may be overestimated. Moreover, consumption data is not available in CHNS, which is widely accepted as better measurement for poverty than income itself as households smooth consumption based on their income flow.

This study, nevertheless, has considerable strengths in using longitudinal data that trace some individuals from early childhood to early adulthood, and can shed some light on the increasing educational inequality in present-day China. In China, the poorest children face multiple challenges that are likely to place them at a competitive disadvantage in the schooling process. For example, in Northwest China, China's poorest region, the poorest children were at elevated risk of being chronically undernourished, living in food insecure households, and lacking access to vision correction (Emily Hannum, Liu, and Frongillo 2012; Emily Hannum and Zhang 2012; Yu and Hannum 2007). Children reported lacking essential school materials and supplies, and fretted constantly about how to get them; they lacked adults at home who had the educational experience to assist them with homework or school problems; and they worried about money and the burden of their school costs on their parents (E. Hannum and Adams 2008). Lack of access to early childhood education for poor rural children also places them at a disadvantage relative to many urban children (Luo et al. 2012). The disproportionate impact of poverty in early and middle childhood on children's long-term educational prospects highlights the potential value of compensatory educational programs targeting early childhood education and the beginning school transition.

#### Acknowledgements

We thank Prof. Jere Behrman and seminar participants at the Population Studies Center at the University of Pennsylvania for comments that improved the paper. This research uses data from China Health and Nutrition Survey (CHNS). We thank the National Institute of Nutrition and Food Safety, China Center for Disease Control and Prevention, Carolina Population Center (5 R24 HD050924), the University of North Carolina at Chapel Hill, the NIH (R01-HD30880, DK056350, R24 HD050924, and R01-HD38700) and the Fogarty International Center, NIH for financial support for the CHNS data collection and analysis files from 1989 to 2011 and future surveys, and the China-Japan Friendship Hospital, Ministry of Health for support for CHNS 2009. The authors thank Grand Challenges Canada Grant 0072-03 for support for undertaking this study.

## 8 **REFERENCES**

Adepoju, Abimbola O. 2012. "Poverty Transitions in Rural South West Nigeria." Global Journal of Science Frontier Research 12 (2). https://globaljournals.org/GJSFR\_Volume12/3-Poverty-Transitions-in-Rural-South-West-Nigeria.pdf.

Alderman, Harold, ed. 2011. No Small Matter: The Impact of Poverty, Shocks, and Human Capital Investments in Early Childhood Development. Washington, D.C.: World Bank Publications. http://siteresources.worldbank.org/EDUCATION/Resources/278200-1298568319076/nosmallmatter.pdf.

Barreca, Alan I. 2010. "The long-term economic impact of in utero and postnatal exposure to malaria." Journal of Human Resources 45.4: 865-892.

Baulch, Bob and John Hoddinott, 2000. Economic mobility and poverty dynamics in developing countries. Journal of Development Studies, Vol 36, No. 6, pp. 1-24

Bayudan-Dacuycuy, Connie, and JosephAnthony Lim. 2014. "Chronic and Transient Poverty and Vulnerability to Poverty in the Philippines: Evidence Using a Simple Spells Approach." Social Indicators Research 118 (1): 389–413. doi:10.1007/s11205-013-0409-5.

Behrman, J. R. *et al.* Nutritional supplementation in girls influences the growth of their children: prospective study in Guatemala. *The American Journal of Clinical Nutrition* **90**, 1372–1379 (2009).

Bird, Kate. 2007. The Intergenerational Transmission of Poverty: An Overview. 99 (October). Chronic Poverty Research Centre. London: Overseas Development Institute. http://papers.ssrn.com/sol3/papers.cfm?abstract\_id=1629262.

Black, Sandra E., Paul J. Devereux, and Kjell G. Salvanes. 2007. "From the Cradle to the Labor Market? The Effect of Birth Weight on Adult Outcomes." The Quarterly journal of economics 122.1: 409-439.

Borgeraas, Elling, and Espen Dahl. 2010. "Low Income and 'poverty Lines' in Norway: A Comparison of Three Concepts." International Journal of Social Welfare 19 (1): 73–83. doi:10.1111/j.1468-2397.2008.00622.x.

Bourguignon, Francois. 2006. "From Income to Endowments: The Difficult Task of Expanding the Income Poverty Paradigm." In Poverty and Inequality, 76–102. Palo Alto, CA: Stanford University Press.

Bourguignon, Francois, and Satya R Chakravarty. 2003. "The Measurement of Multidimensional Poverty." Journal of Economic Inequality 1 (1): 25–49.

Boyden, Jo, and Elizabeth Cooper. 2007. Questioning the Power of Resilience: Are Children up to the Task of Disrupting the Transmission of Poverty? Working Paper 73 (April 1). Chronic Poverty Research Centre. London: Overseas Development Institute. http://dx.doi.org/10.2139/ssrn.1753009.

Bradbury, Bruce, Stephen Jenkins and John Micklewright, 2001. The Dynamics of Child Poverty in Industrialized Countries. Cambridge University Press, UK

Brady, David. 2003. "Rethinking the Sociological Measurement of Poverty." Social Forces 81 (3): 715–51. doi:10.2307/3598174.

Brooks-Gunn, Jeanne, and Greg J. Duncan. 1997. "The Effects of Poverty on Children." The Future of Children 7 (2): 55–71. doi:10.2307/1602387.

Chen, Zhihong, 2006. "Measuring the poverty lines for urban households in China an equivalence scale method." China Economic Review 17.3: 239-252.

Cherng, Hua-Yu Sebastian, and Emily Hannum. 2013. "Community Poverty, Industrialization, and Educational Gender Gaps in Rural China." Social Forces 92 (2): 659– 90. doi:10.1093/sf/sot084.

Chiu, Ming Ming. 2007. "Families, Economies, Cultures, and Science Achievement in 41 Countries: Country-, School-, and Student-Level Analyses." Journal of Family Psychology 21 (3): 510–19. doi:10.1037/0893-3200.21.3.510.

Christiaensen, Luc, and Anthony Shorrocks. 2012. "Measuring Poverty over Time." The Journal of Economic Inequality 10 (2): 137–43. doi:10.1007/s10888-012-9225-4.

Crookston, Benjamin T, Whitney Schott, Santiago Cueto, Kirk A Dearden, Patrice Engle, Andreas Georgiadis, Elizabeth A Lundeen, Mary E Penny, Aryeh D Stein and Jere R Behrman. 2013. "Postinfancy Growth, Schooling, and Cognitive Achievement: Young Lives." The American Journal of Clinical Nutrition, 98(6), 1555-63,

Davis, Deborah, Pierre Landry, Yusheng Peng, and Jin Xiao. 2007. "Gendered Pathways to Rural Schooling: The Interplay of Wealth and Local Institutions." China Quarterly 189: 60–60–82.

Deaton, Angus. 1997. "The Analysis of Household Surveys: A Microeconometric Approach to Development Policy." Baltimore and London: The Johns Hopkins University Press for the World Bank.

Dercon, S. 2012. "Understanding Child Poverty in Developing Countries: Measurement and Analysis." In Childhood Poverty: Multidisciplinary Approaches, edited by J. Boyden and M. Bourdillon, 52–72. New York: Palgrave Macmillan.

Duncan, Greg J, and Katherine Magnuson. 2013. "The Long Reach of Early Childhood Poverty." In Economic Stress, Human Capital, and Families in Asia, 57–70. Springer.

Duncan, Greg J., W. Jean Yeung, Jeanne Brooks-Gunn, and Judith R. Smith. 1998. "How Much Does Childhood Poverty Affect the Life Chances of Children?" American Sociological Review 63 (3): 406–23. doi:10.2307/2657556.

Engle, Patrice L, and Maureen M Black. 2008. "The Effect of Poverty on Child Development and Educational Outcomes." Annals of the New York Academy of Sciences 1136 (1): 243–56.

Engle, Patrice L. 2012. "Poverty and Developmental Potential." In Childhood Poverty: Multidisciplinary Approaches, edited by J. Boyden and M. Bourdillon, 129–47. New York: Palgrave Macmillan.

Filmer, Deon, and Lant Pritchett. 1999. "The Effect of Household Wealth on Educational Attainment: Evidence from 35 Countries." Population and Development Review 25 (1): 85–120.

Foster, James E. 1998. "Absolute versus Relative Poverty." The American Economic Review 88 (2): 335–41. doi:10.2307/116944.

Foster, James, Joel Greer, and Erik Thorbecke. 1984. "A Class of Decomposable Poverty Measures." Econometrica 52 (3): 761–66. doi:10.2307/1913475.

Gershoff, Elizabeth T., J. Lawrence Aber, C. Cybele Raver, and Mary Clare Lennon. 2007. "Income Is Not Enough: Incorporating Material Hardship Into Models of Income Associations With Parenting and Child Development." Child Development 78 (1): 70–95. doi:10.1111/j.1467-8624.2007.00986.x.

Grant, Monica J., and Jere R. Behrman. 2010. "Gender Gaps in Educational Attainment in Less Developed Countries." Population and Development Review 36 (1): 71–89. doi:10.1111/j.1728-4457.2010.00318.x.

Grantham-McGregor, Sally, Yin Bun Cheung, Santiago Cueto, Paul Glewwe, Linda Richter, and Barbara Strupp. 2007. "Developmental Potential in the First 5 Years for Children in Developing Countries." The Lancet 369 (9555): 60–70.

Grusky, David B., and Ravi Kanbur. 2006. "The Conceptual Foundations of Poverty and Inequality Measurement." In Poverty and Inequality, edited by David B. Grusky and Ravi Kanbur, 1–29. Stanford, CA: Stanford University Press. http://mammoth.trentu.ca/ids/documents/Q2\_WP21\_GruskyandKanbur.pdf.

Guo, Guang. 1998. "The Timing of the Influences of Cumulative Poverty on Children's Cognitive Ability and Achievement." Social Forces 77 (1): 257–87. doi:10.2307/3006017.

Hannum, E., and Jennifer Adams. 2008. "Beyond Cost: Rural Perspectives on Barriersto Education." In , edited by D. Davis and F. Wang, 156–71. Creating Wealth and Poverty inChina.Stanford,CA:StanfordUniversityPress.http://repository.upenn.edu/cgi/viewcontent.cgi?article=1006&context=gansu\_papers.

Hannum, Emily. 2003. "Poverty and Basic Education in Rural China: Villages, Households, and Girls' and Boys' Enrollment." Comparative Education Review 47 (2): 141– 59. doi:10.1086/376542.

Hannum, Emily, Jihong Liu, and Edward A. Frongillo. 2012. "Poverty, Food Insecurity and Nutritional Deprivation in Rural China: Implications for Children's Literacy Achievement." International Journal of Educational Development, no. available on-line 13 September 2012. doi:10.1016/j.ijedudev.2012.07.003. Hannum, Emily, and Yuping Zhang. 2012. "Poverty and Proximate Barriers to Learning: Vision Deficiencies, Vision Correction and Educational Outcomes in Rural Northwest China." World Development 40 (9): 1921–31. doi:10.1016/j.worlddev.2012.04.029.

Hanushek, E.A., and L. Woessmann. 2011. The Economics of International Differences in Educational Achievement. Vol. 3. http://www.scopus.com/inward/record.url?eid=2-s2.0-79251594612&partnerID=40&md5=51807331adb4c33ca1d19e0583f39c6d.

Haughton, Jonathan Henry, and Shahidur R. Khandker. 2009. Handbook on Poverty and Inequality. Washington, DC: World Bank. http://go.worldbank.org/7JGPK76TM0.

Hoddinott, John and Bill Kinsey, 2001. "Child Growth in the Time of Drought". Oxford Bulletin of Economics and Statistics, Vol 63, No. 4, pp. 409-436

Hoddinott, John F., John A. Maluccio, Jere R. Behrman, Rafael Flores and Reynaldo Martorell. 2008 "Effect of a Nutrition Intervention During Early Childhood on Economic Productivity in Guatemalan Adults." Lancet, 371(9610), 411-16.

Holmes, John, and Kathleen Kiernan. 2013. "Persistent Poverty and Children's Development in the Early Years of Childhood." Policy & Politics 41 (1): 19–42.

Huisman, Janine, and Jeroen Smits. 2009. "Effects of Household- and District-Level Factors on Primary School Enrollment in 30 Developing Countries." World Development 37 (1): 179–93. doi:10.1016/j.worlddev.2008.01.007.

Jones-Smith, Jessica C., and Barry M. Popkin. 2010. "Understanding community context and adult health changes in China: development of an urbanicity scale." Social science & medicine 71 (8): 1436-1446.

Ladd, Helen F. 2012. "Education and Poverty: Confronting the Evidence." Journal of Policy Analysis and Management 31 (2): 203–27. doi:10.1002/pam.21615.

Luo, Renfu, Linxiu Zhang, Chengfang Liu, Scott Rozelle, Brian Sharbono, and Jennifer Adams. 2012. "Behind Before They Begin: The Challenge of Early Childhood Education in Rural China." Australasian Journal of Early Childhood 37 (1).

Maluccio, John A., John F. Hoddinott, Jere R. Behrman, Agnes R. Quisumbing, Reynaldo Martorell and Aryeh D. Stein. 2009. "The Impact of Improving Nutrition During Early Childhood on Education among Guatemalan Adults." Economic Journal, 119(537), 734-63.

Marlier, Eric, and A. B. Atkinson. 2010. "Indicators of Poverty and Social Exclusion in a Global Context." Journal of Policy Analysis and Management 29 (2): 285–304. doi:10.1002/pam.20492.

Nussbaum, Martha M. 2006. "Poverty and Human Functioning: Capabilities as Fundamental Entitlements." In Poverty and Inequality, edited by David B. Grusky and Ravi Kanbur, 47–75. Stanford, CA: Stanford University Press. Pells, Kirrily. 2011. Poverty and Gender Inequalities: Evidence from Young Lives. Oxford University Department of International Development Young Lives Policy Paper 3. Oxford: Department of International Development. http://r4d.dfid.gov.uk/pdf/outputs/younglives/yl\_pp3\_poverty-and-gender-inequalities.pdf.

Popkin, Barry M. 1994. "Social Change and Its Nutritional Impact: The China Health and Nutrition Survey." China Exchange News 22 (2): 9–11.

Prentice, Andrew M, Kate A Ward, Gail R Goldberg, Landing M Jarjou, Sophie E Moore, Anthony J Fulford and Ann Prentice. 2013. "Critical Windows for Nutritional Interventions against Stunting." The American Journal of Clinical Nutrition, *97*(5), 911-918.

Ravallion, Martin, and Shaohua Chen. 2007. "China's (uneven) Progress against Poverty." Journal of Development Economics 82 (1): 1–42.

Royer, Heather. 2009. "Separated at girth: US twin estimates of the effects of birth weight." American Economic Journal: Applied Economics 1.1: 49-85.

Scott, Christopher D. 2000. "Mixed fortunes: a study of poverty mobility among small farm households in Chile, 1968–86." The Journal of Development Studies 36.6: 155-180.

Sen, Amartya. 1999. Commodities and Capabilities. Delhi; New York: Oxford University Press. http://EconPapers.repec.org/RePEc:oxp:obooks:9780195650389.

——. 2006. "Conceptualizing and Measuring Poverty." In Poverty and Inequality, edited by David B. Grusky and Ravi Kanbur, 30–46. Stanford, CA: Stanford University Press.

Schott, Whitney, Benjamin T. Crookston, Elizabeth A. Lundeen, Aryeh D. Stein, Jere R. Behrman and Young Lives Determinants and Consequences of Child Growth Project Team. 2013. "Child Growth from Ages 1 to 8 Years in Ethiopia, India, Peru and Vietnam: Key Distal Household and Community Factors." Social Science & Medicine, 97, 278-87.

Singh, Renu, and Sudipa Sarkar. 2014. "Children's Experience of Multidimensional Deprivation: Relationship with Household Monetary Poverty." The Quarterly Review of Economics and Finance, July. doi:10.1016/j.qref.2014.06.007.

Thorbecke, Erik. 2005. Multidimensional Poverty: Conceptual and Measurement Issues. Conference Paper. Brasilia: UNDP International Poverty Centre International Conference on "The Many Dimensions of Poverty", August 29-31. http://econweb.arts.cornell.edu/et17/Erik%20Thorbecke%20files/Multi1.pdf.

Van der Gaag, Nikki, Kirrily Pells, and Caroline Knowles. 2012. Changing Lives in a Changing World: Young Lives Children Growing up. Oxford: University of Oxford Department for International Development. http://www.younglives.org.uk/publications/B/changing-lives-in-a-changing-world.

Walker, Susan P, Theodore D Wachs, Sally Grantham-McGregor, Maureen M Black, Charles A Nelson, Sandra L Huffman, Helen Baker-Henningham, et al. 2011. "Inequality in Early Childhood: Risk and Protective Factors for Early Child Development." The Lancet 378 (9799): 1325–38. World Bank. 2011. "Poverty Analysis - Measuring Poverty". Education and Training. Poverty Reduction and Equity. http://go.worldbank.org/0C60K5UK40.

Wu, Xiaogang, and Zhuoni Zhang. 2010. "Changes in Educational Inequality in China, 1990–2005: Evidence from the Population Census Data." In , edited by Emily Hannum, Hyunjoon Park, and Yuko G. Butler, 17:123–52. Globalization, Changing Demographics, and Educational Challenges in East Asia. Emerald Group Publishing Limited. http://dx.doi.org/10.1108/S1479-3539(2010)0000017007.

Victora, Cesar G, Linda Adair, Caroline Fall, Pedro C Hallal, Reynaldo Martorell, Linda Richter, Harshpal Singh Sachdev and Maternal and Child Undernutrition Study Group. 2008. "Maternal and Child Undernutrition: Consequences for Adult Health and Human Capital." The Lancet, 371(9609), 340-57.

Victora, Cesar Gomes, Mercedes de Onis, Pedro Curi Hallal, Monika Blössner and Roger Shrimpton. 2010. "Worldwide Timing of Growth Faltering: Revisiting Implications for Interventions." Pediatrics, peds-2009

Yoshikawa, Hirokazu, J. Lawrence Aber, and William R. Beardslee. 2012. "The Effects of Poverty on the Mental, Emotional, and Behavioral Health of Children and Youth: Implications for Prevention." American Psychologist 67 (4): 272–84. doi:10.1037/a0028015.

Yu, Shengchao, and Emily Hannum. 2007. "Food for Thought: Poverty, Family Nutritional Environment, and Children's Educational Performance in Rural China." Sociological Perspectives 50 (1): 53–77.

# APPENDIX

Dependent variables: hazard ratio of dropping out of formal schooling									
	(1)	(2)	(3)	(4)	(5)				
(Log) long time income mean	-0.336***		-0.335**		-0.336**				
Poor at age<4 <sup>a</sup>	(0.111)	0.11	0.0446	0.353**	(0.154) 0.291*				
Poor at age [4,10) <sup>b</sup>		(0.107) 0.287***	(0.114) 0.194*	(0.155) 0.368**	(0.164) 0.28				
Poor at age [10,16) <sup>c</sup>		(0.109) -0.029	(0.115) -0.177	(0.174) -0.14	(0.184) -0.261				
Male * poor at age<4 <sup>d</sup>		(0.123)	(0.139)	(0.172) -0.397**	(0.180) -0.399**				
Male * poor at age [4,10) <sup>e</sup>				(0.196) -0.117	(0.202) -0.127				
Male * poor at age [10.16) <sup>f</sup>				(0.213) 0.172	(0.222) 0.125				
Male	0 0922	0 0988	0 0943	(0.227) 0.332*	(0.234) 0.348*				
	(0.098)	(0.100)	(0.100)	(0.183)	(0.183)				
Prob > F test of (a + d=0)				0.063	0.124				
Prob > F test of (b + $e=0$ )				0.023	0.198				
Prod > F  test of  (C + f=0)				0.691	0.310				
Log pseudolikelihood	-599.242	-599.256	-597.698	-597.891	-596.343				
Note: The table shows the survival analy	sis on finished	schooling in a	dulthood by fitt	ing Cox propo	rtional				

### Table A.1. Poverty spell and education attainment at adulthood (n=1,058)

Note: The table shows the survival analysis on finished schooling in adulthood by fitting Cox proportional hazard model with censoring. The hazards are stratified by province, urban/rural area and birth year. All specifications control for parents' education level dummies, quadratic function of age, number of children under 16 in the household at each growth period, urbanization index of the community during the whole childhood, and urbanization index for education. Robust standard errors clustered at the community level are reported in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Dependent variables: hazard ratio of dropping out of formal schooling							
	(1)	(2)	(3)	(4)			
Chronic poverty index <sup>a</sup>	-1.276 (1.893)	-3.99 (5.647)	-1.599 (1.893)	-4.274 (5.650)			
Transient poverty index <sup>b</sup>	2.408*** (0.597)	3.297*** (0.767)	<b>、</b>	, , ,			
Chronic poverty* male <sup>c</sup>		3.675 (6.033)		3.584 (5.961)			
Transient poverty * male <sup>d</sup>		-1.738* (0.889)					
Transient poverty index at age [0,4) <sup>e</sup>			0.529** (0.212)	1.388*** (0.301)			
Transient poverty index at age [4,10) <sup>f</sup>			0.557*** (0.204)	0.14 (0.347)			
Transient poverty index at age [10,16) <sup>g</sup>			0.225 (0.211)	0.615* (0.327)			
Transient poverty index at age [0,4) * male <sup>h</sup>				-1.386*** (0.360)			
Transient poverty index at age [4,10) * male <sup>i</sup>				0.748* (0.436)			
Transient poverty index at age [10,16) * male <sup>j</sup>				-0.521 (0.458)			
Male	0.0946 (0.098)	0.208 (0.133)	0.0852 (0.099)	0.231* (0.140)			
Prob > F test of (a+c=0) Prob > F test of (b+d=0)		0.770 0.000		0.707			
Prob > F test of (e+h=0)				0.000			
Prob > F test of (f+i=0)				0.002			
Prob > F test of (g+j=0)				0.158			
Log pseudolikelihood	- 597.127	- 596.266	- 597.940	- 593.061			

Table A.2. Poverty persistence and education attainment at adulthood (n=1,058)

Note: The table shows the survival analysis on finished schooling in adulthood by fitting Cox proportional hazard model with censoring. The hazards are stratified by province, urban/rural area and birthyears. All specifications control for parents' education level dummies, quadratic function of age, number of children under 16 in the households during each growth period, urbanization index of the community during the whole childhood, and urbanization index for education. Robust standard errors clustered at community level are reported in parentheses. \*\*\* p<0.01, \*\* p<0.05, \* p<0.1